Author's response to reviews

Title: Acute abdominal compartment syndrome complicating a colonoscopic perforation: a case report.

Authors:

amine souadka (souadka@gmail.com)
raouf Mohsine (pr.mohsine@gmail.com)
lahsen Ifrine (oelmalki@hotmail.fr)
Abdelkader Belkouchi (belkouchi@gmail.com)
Hadj Omar El Malki (oelmalki@gmail.com)

Version: 4 Date: 10 January 2012

Author's response to reviews: see over
10th January, 2012

to the deputy editor of Journal of Medical Case Report

Dear Editor,

Title: Acute abdominal compartment syndrome complicating a colonoscopic perforation: report of case.

I would like to inform you that the changes requested by the editorial team, concerning the ethnicity of the patient both in the abstract and case presentation section of the manuscript were corrected.

In the abstract: north-african was added at the second line of the case presentation.

In the manuscript section: north-african was also added at the first line of the case presentation.

Looking forward to hearing from you soon

Sincerely yours,

Dr Souadka Amine
Surgical department A
Ibn Sina Hostpital
Rabat, Morocco

Email: souadka@gmail.com
Phone number: +212 6 66 95 36 68